Author's response to reviews

Title: Primary growth hormone insensitivity (Laron syndrome) and acquired hypothyroidism: a case report

Authors:

Oana R Cotta (oanacotta@yahoo.gr)
Libero Santarpia (lsantarp@mdanderson.org)
Lorenzo Curto (curto.loren@libero.it)
Gianluca Aimaretti (gianluca.aimaretti@med.unipmn.it)
Ginevra Corneli (endocrinologia@unime.it)
Francesco Trimarchi (Francesco.Trimarchi@unime.it)
Salvatore Cannavò (cannavos@unime.it)

Version: 5 Date: 7 December 2010

Author's response to reviews:

Major revisions
Title has been changed to: ‘PRIMARY GROWTH HORMONE INSENSITIVITY (LARON SYNDROME) AND ACQUIRED HYPOTHYROIDISM: A CASE REPORT’
The patient’s ethnicity has been included in the Abstract -case presentation- section of the manuscript.

Minor revisions
All highlighting has been removed from the manuscript.
The heading 'Patient Presentation' has been changed to 'Case Presentation'.
Keywords have been removed.
The manuscript has been checked for typographical errors and they have all been corrected.